Title: Bipolar patient treated with lithium, haloperidol and benztropine during pregnancy presented with unexpected depletion in plasma choline and phosphatidylcholine: a case report

Authors:

Maxine Gossell-Williams (maxine.gossell@uwimona.edu.jm)
Horace Fletcher (horace.fletcher@uwimona.edu.jm)
Steven Zeisel (steven_zeisel@unc.edu)

Version: 5 Date: 14 January 2008

Author's response to reviews: see over
January 10, 2008

Editor-in-chief
Journal Medical Case Reports

Dear Sir/Madam,

In response to your letter dated January 9, 2008 please find attached the re-submitted article (MS: 6830416771600629) titled :
“Bipolar patient treated with lithium, haloperidol and benztropine during pregnancy presented with unexpected depletion in plasma choline and phosphatidylcholine: a case report”

for consideration for publication in “Journal Medical Case Reports”.

The paper has been revised, addressing concerns with grammar. Below is a list of the changes made:

1. Page 3, paragraph 1, line 1: “the” was removed from in front of “phosphatidylcholine”.
2. Page 3, paragraph 2, line 2: the word “studies” has been added.
3. Page 3, paragraph 3: comma added after “category D” and “which include” was changed to “including”.
4. Page 5, paragraph 1, line 1: “Jamaica” was changed to “Jamaican”
5. Page 5, paragraph 1, line 2: reworded to reflect “no follow-up data”
6. Page 5, paragraph 2, line 6: the word “bipolar” was added.
7. Page 6, paragraph 3, line 4: “decrease” was changed to “decreases” and “was” changed to “were”.
8. Page 6, paragraph 3, line 6: “possible” changed to “possibly”.
9. Page 7, consent subheading, line 1: the word “the” was removed.

Regards,
Maxine Gossell-Williams